Author's response to reviews

Title: Down syndrome with microgranular variant of acute promyelocytic leukemia in a child - A case report

Authors:

Deepali Jain (deepalijain76@gmail.com)
Tejindar Singh (tsinghmamc@yahoo.co.in)
Prerna Arora (parora@rediffmail.com)

Version: 2 Date: 8 September 2007

Author's response to reviews: see over
To,

Editor-in-Chief

Journal of medical case report

Subject: Submission of manuscript entitled “Down syndrome with microgranular variant of acute promyelocytic leukemia in a child - A case report”

Dear Sir,

Kindly find enclosed manuscript entitled “Down syndrome with microgranular variant of acute promyelocytic leukemia in a child - A case report” for consideration of publication in your esteemed journal. [1] All authors of article have made a substantial contribution to the information or material submitted for publication; [2] All have read and approved the final manuscript; [3] The manuscript or portion of it is not under consideration by any other journal and have not been previously published. We do hope that the article would be of interest to Journal of medical case report. We request you to kindly acknowledge the receipt of the manuscript along with enclosures.

Thanking you
Yours sincerely,

Dr. Deepali Jain, M.D.DNB

Assistant professor

Department of Pathology

Maulana azad medical college

New Delhi, India 110002

Email : deepalijain76@gmail.com

Phone : 91-98688-95112